Contents lists available at ScienceDirect

Energy Reports

journal homepage: www.elsevier.com/locate/egyr



Research paper

Exploring the roles of storage technologies in the Spanish electricity system with high share of renewable energy

Sébastien Huclin^{a,b,*}, José Pablo Chaves^a, Andrés Ramos^a, Michel Rivier^a, Teresa Freire-Barceló^a, Francisco Martín-Martínez^a, Tomás Gómez San Román^a, Álvaro Sánchez Miralles^a

^a Institute for Research in Technology (IIT)-ICAI School of Engineering, Universidad Pontificia Comillas, Santa Cruz de Marcenado 26, 28015, Madrid, Spain Berner, Curter Grant, Charge (BC2), Sede Building 1, 1st George Scientific Commun. 40040, Leize Scient

^b Basque Centre for Climate Change (BC3), Sede Building 1, 1st floor, Scientific Campus, 48940, Leioa, Spain

ARTICLE INFO

Article history: Received 15 September 2021 Received in revised form 11 February 2022 Accepted 6 March 2022 Available online 23 March 2022

Keywords: Renewables Electricity Critical events Hydropower Operation planning Storage

ABSTRACT

At operational level, fossil fuel phase-out and high shares of non-dispatchable renewable energy resources (RES) will challenge the system operator's (SO) ability to balance generation, and the demand at any time. The variability of RES output ranges from one hour to a season, and critical events such as low supply and high demand might occur more frequently and for more extended periods. When evaluating the role of Energy Storage Systems (ESSs) in this context, the need for a long time scope to capture the different RES variabilities must be reconciled with the need for modeling the hourly chronology. This paper presents a medium-term operation planning model, addressing both the energy dispatch and the balancing services. This study shows that representing the combined chronological variability of demand and RES production is essential to properly assess the roles of different kinds of ESSs in the future 2030 electricity mix. Otherwise, it would not be possible to appropriately capture the frequency, depth, and length of events for which ESSs are activated. The analysis also highlights the importance of considering balancing services, given the significant contribution of batteries to the reserve market. Finally, the results show that batteries and Pumped Storage Hydro (PSH) have different roles in the Spanish electricity system with a high renewable penetration. While PSH is mainly used to provide energy during critical periods, batteries mostly provide balancing services.

© 2022 The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

1. Introduction

The fossil fuel phase-out and high shares of non-dispatchable renewable energy resources (RES) such as solar and wind will challenge the operation of power systems (MITECO, 2021). The variability and the non-dispatchable nature of RES may jeopardize the security of supply and the provision of operating reserves. Energy Storage Systems (ESSs) such as Pumped Storage Hydro units (PSHs) and batteries are technologies that can shift energy at different timeframes to cope with the high variability of RES production. Thus, ESS could help reduce renewable curtailment and provide security of supply in renewable-dominated power systems (Qadir et al., 2021).

Although PSHs have operated for a long time in power systems, the current existing installed capacity is insufficient, and installing new ESSs, such as batteries, could help meet renewable

* Corresponding author at: Institute for Research in Technology (IIT)-ICAI School of Engineering, Universidad Pontificia Comillas, Santa Cruz de Marcenado 26, 28015, Madrid, Spain.

E-mail address: shuclin@comillas.edu (S. Huclin).

targets by 2030. Since batteries differ from PSHs in their technical and economic parameters (Shaqsi et al., 2020), it is helpful to know how complementary and competitive they are in operational terms. For instance, batteries have shorter charge cycles and higher roundtrip efficiency than PSHs (Mongird et al., 2019).

In the context of the energy transition, Spain is an interesting real-case to study the challenges of integrating non-dispatchable RESs with battery and PSH technologies since the Spanish National Energy and Climate Plan (NECP) (MITECO, 2021) aims for a minimum of 74% of its electricity to be produced by renewables by 2030. Between 2020 and 2030, the Spanish NECP foresees a significant installed capacity increase of 70% for solar photovoltaic (Solar PV), 40% for wind, and 64% for ESS - i.e., PSH and batteries. The future ESS mix in Spain will consist of batteries and new and existing PSHs (BOE, 2021). Additionally, the Climate Change and Energy Transition Project Law foresees that the operation of all available ESS will contribute to the security and continuity of the electricity system at all times (BOE, 2021). Indeed, battery participation in energy and ancillary services was included in December 2020 in the Spanish System Operator (SO) operating procedures (BOE, 2020).

https://doi.org/10.1016/j.egyr.2022.03.032

2352-4847/© 2022 The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).



Abbreviations			$ au_s$
BC	Base case		
CCGT	Combined Cycle Gas Turbine		62,00
CLPSH	Closed-Loop Pumped Storage Hydro		
ESS	Energy Storage System		
MIP	Mixed Integer Linear Programming		$DR_{\omega,n}, UR_{\omega,n}$
NECP	National Energy and Climate Plan		۵,۳۰ ۵,۳
OLPSH	Open-Loop Pumped Storage Hydro		RD_t, RU_t
PCI	Project of Common Interest		
PSH	Pumped Storage Hydro		TD_t, TU_t
REE	Red Eléctrica de España (Spanish system		
DEC	operator) Popowable Energy Sources		CSD_g, CSU_g
RES	Run of River bydro power plant		
SO	System Operator		Variables
Solar DV	Solar photovoltaic		ens
	Solar processors and the solar solar solar processors and the solar so		gnon a gCon
UGII	de Gestion Hidráulica in Spanish)		<i>Gr</i> <i>w</i> , <i>n</i> , <i>g</i> , <i>G</i> - <i>w</i>,<i>n</i>
Nomenclature			$p_{\omega.n,g}$
Indices			Cons
(i)	Scenario		- <i>w</i> , <i>n</i> ,s
n	Load Level		$uc_{\omega,n,g}$
σ	All kind of generating unit (thermal		,
8	hydro, RES and ESS unit)		$sd_{\omega,n,g}, su_{\omega,n}$
t	Thermal unit		
h	Hydro unit		$arphi_{\omega,n,or}$
or	Unit able to provide operating reserve		
S	Energy Storage System (ESS)		6.6
r	Energy reservoir associated with an ESS		$SOC_{\omega,n,s}$
Parameters			$S_{\omega,n,s}$
D _{on} n	Hourly load demand [MW]		_
DUR,	Duration of each load level [h]		$dr_{\omega,n,or}, ur_{\omega,n}$
CENS	Cost of Energy Non-Served [€/MWh]		
P	Probability of each scenario [p.u.]		da/ sun/
EFs	Roundtrip efficiency for ESS unit		$u_{\omega,n,s}, u_{\omega,n,s}$
5	(i.e., percentage of electricity put into		ad au
	storage that is latter retrieved) [p.u.]		$uu_{\omega,n,or}, uu_{\omega,n}$
$EI_{\omega,n,s}$	Weekly energy inflow for ESS unit		
	[MWh]		
$IUC_{\omega,g}$	Initial unit commitment status for the		$ad'_{\omega,n,s}, au'_{\omega,n}$
	first hour of the year {0–1}		
$P_g, \underline{P_g}$	Maximum and minimum output of gen-		
\overline{C}	Maximum numning of FSS unit [MW]	l	
C_{s}	Total variable cost of generating unit		
CVg	(includes fuel, variable O&M and emis-	1	tosa et al., 200
	sion costs) [€/MWh]]	ower thermal
CV_{s}	Variable cost of ESS unit [€/MWh]	i	age units with
$R_{\omega n}$	Ratio between upward reserves and to-	1	imeframes. Inc
<i>w</i> , <i>n</i>	tal operating reserves set by the SO	i	ntlows require
	[p.u.]	(count for the h
Is	Energy capacity of ESS unit [MWh]	(In nower sy
			in power sy

Traditionally, hydrothermal scheduling, considering energy and operating reserves, has been studied through models that reproduce the yearly operation on an hourly basis and minimize the system's total cost under different operational constraints (Ven-

$ au_s$	Duration of ESS discharge cycle (e.g., 24, 168, 672 for daily, weekly, monthly) [h]
CD, CU	Percentage use of energy concerning the balancing capacity for generating unit providing downward/upward operating reserves [%]
$DR_{\omega,n}$, $UR_{\omega,n}$	Hourly downward and upward operat- ing reserve [MW]
RD_t, RU_t	Ramp down and ramp up limits of thermal unit [MW/h]
TD_t, TU_t	Minimum downtime and uptime of thermal unit [h]
CSD_g, CSU_g	Shutdown and start-up and costs of committed unit $[M \\mathcal{E}]$
Variables	
$ens_{\omega,n}$	Energy non-served [MWh]
$gp_{\omega,n,g}, gc_{\omega,n,g}$	Generator output and consumption (dis- charge if ESS) [MW]
$p_{\omega.n,g}$	Production of unit above minimum out- put [MW]
$C_{\omega,n,s}$	Consumption of ESS unit above mini- mum output [MW]
$\mathcal{UC}_{\omega,n,g}$	Commitment of a generation unit per load level {0,1}
$sd_{\omega,n,g}, su_{\omega,n,g}$	Shutdown, and start-up event of gener- ating unit {0,1}
$arphi_{\omega, {\sf n}, {\sf or}}$	Commitment in activating balancing en- ergy from balancing capacity provided {0,1}
$SoC_{\omega,n,s}$	State of Charge (SoC) of the energy reservoir for ESS unit [MWh]
$S_{\omega,n,s}$	Spilled energy of the energy reservoir for ESS unit [MWh]
$dr_{\omega,n,or}, ur_{\omega,n,or}$	Provision of upward and downward operating reserves for generating unit [MW]
$dr'_{\omega,n,s}, ur'_{\omega,n,s}$	Provision of upward and downward operating reserves for ESS [MW]
$ad_{\omega,n,or}, au_{\omega,n,or}$	Activated energy associated to the pro- vision of upward and downward op- erating reserves for generating unit [MWh]
$ad'_{\omega,n,s}, au'_{\omega,n,s}$	Activated energy associated to the pro- vision of upward and downward operat- ing reserves for ESS [MWh]

tosa et al., 2005). Such models need to be adapted to address lower thermal generation scenarios, and to represent other storage units with different characteristics, and charge and discharge timeframes. Indeed, the variability of RES production and water inflows requires considering a wide range of timeframes to account for the hourly, daily, weekly, and even seasonal variability of these resources.

In power systems with large share of non-dispatchable RES, periods with low non-dispatchable RES production are becoming significantly challenging. In the Ten-Year Network Development Plan (TYNDP) 2022 Scenario Building Guidelines (ENTSO-E, 2021a), ENTSO-E defines 'DunkelFlaute' scenarios as the two weeks of the year with high demand combined with low solar and wind generation. Although critical events often refer to high load demand and low non-dispatchable RES production, critical events

should also be defined according to thermal unit unavailability. Since non-dispatchable RES production shares will increase, the frequency and duration of critical events are also expected to grow. Therefore, a medium-term scope – i.e., considering intertemporal constraints from months to hours – respecting the chronology of events seems necessary when studying the operating behavior of the different ESSs in RES-dominated power systems. Furthermore, future RES-dominated power systems with a high impact of forecast errors require analysis of the effects of the future ESS mix on system operation, including their role in energy arbitrage and the provision of auxiliary services. This paper specifically analyzes operating reserves.

The method must be technology-neutral when comparing ESS with different charge cycles. According to the time dimension, analysis of the roles of different ESSs requires a sufficiently long time scope and a time scope short enough to represent the operation of batteries and PSH accurately. Too short a time scope (<1 year) would not allow consideration of the seasonal variability of water inflows. While too long a time scope (>1 h) would not allow a correct picture of the operation of the batteries, given that the latter presents a storage capacity equivalent to a few hours. Additionally, the methodology must consider operating reserve since battery technology is authorized to provide balancing services in the context of the security of supply.

This paper aims to analyze the role of batteries and PSHs in the future ESS mix by exploring the 2030 Spanish NECP scenario. Due to its relevance for this study, the annual operation of the whole system is modeled in detail, with hourly temporal granularity. The model also optimizes the unit commitment of the thermal generating units and considers RES hourly data sets that allow accounting for the variability of wind, solar, and hydropower inflows. For hydropower units, the model considers weekly inflows for market-cleared generating units. The model maintains consistency and chronology between all input data. Particular attention is given to the representation of different ESS units with different storage timeframes in a medium-term operation planning model. The system operation is formulated as a large-scale mixed-integer programming problem (MIP).

Additionally, the model considers both the supply of energy and balancing services, including operating reserve capacity – i.e., balancing capacity – and activating such capacity -i.e., balancing energy-. The study explores different scenarios and sensitivities to account for the effects of critical parameters in the system operation. Under these different scenarios, the competition, and complementarities in operational terms among different ESSs are studied. The results also show how ESS technologies behave during critical hours of power systems.

This article is structured as follows: Section 2 reviews the modeling of ESS impacts in the electricity system and highlights the main challenges to be considered. Section 3 describes the methodology applied and the model considered. Section 4 presents the case study considered. Section 5 details the scenarios explored, and results are provided and analyzed in Section 6. Section 7 summarizes the main conclusions.

2. State of the art

Maintaining the security and continuity of the electricity supply is a fundamental task for the SO, and it has several mechanisms at its disposal, including balancing services (The European Commission, 2017). Frequency regulation services allow the SO to maintain frequency stability and energy balance. These services, also known as operating reserves, have different activation time horizons, ranging from a few seconds to two hours. These services adjust energy injections and withdrawals, upwards and downwards respectively, to cope with forecast errors from generation and demand (The European Commission, 2017), contingencies and forced outages from generation and network assets. Generally, there are two secondary and tertiary frequency regulation products: availability and activation (The European Commission, 2017). In other words, these services are offered firstly in terms of capacity (i.e., MW) and secondly in terms of energy (i.e., MWh).

According to Heuberger et al. (2017), as non-dispatchable RES integration increases, the amount of required operating reserves will increase, too. Furthermore, in REE (2009) the Spanish SO details the role of PSH in electricity generation and shows that hydropower technology (i.e., dams and PSHs) is dominant in the supply of operating reserves. Moreover, Diaz et al. (2014) informs us that the integration of non-dispatchable RESs and the value of the PSH are even more critical when the PSH participates strongly in the provision of operating reserves.

In Oyler and Parsons (2020), the authors focused on how investments in PSH improve greenhouse gas mitigation in scenarios with a high share of non-dispatchable RES according to Distributed Generation scenario 2030 of the TYNDP (ENTSO-E, 0000a). They use a hydrothermal linear medium-term operation planning model with hourly operation details for representing the Spanish electricity system. The authors demonstrated the important role that PSH will have when increasing non-dispatchable RES shares. Nevertheless, the methodology used in Oyler and Parsons (2020) does not consider operating reserves. Furthermore, the scenarios analyzed in Oyler and Parsons (2020) do not consider the projection of the Spanish National Energy and Climate Plan (NECP) (MITECO, 2021) in terms of the increase of PSHs in Spain for 2030, which is expected to increase 50%. As a result, the replicated system does not have the same generation flexibility as the one foreseen by the NECP scenario. Therefore, although the analysis provided in Oyler and Parsons (2020) brings essential insights around the interactions between investment in PSHs and the share of non-dispatchable RES, according to Heuberger et al. (2017), REE (2009) and Diaz et al. (2014), the assessment of the role of ESSs in scenarios with high RES is not explored since the applied methodology is not technology-neutral.

Similar to Oyler and Parsons (2020), the authors in Dollinger and Dietrich (2013) focused on how ESS can help increase wind and solar energy shares in the Spanish electricity mix by exploring 2020 scenarios. Authors have used a hydrothermal mediumterm operation planning model with hourly operation details accounting for energy and operating reserves detailed in Dietrich et al. (2012). Authors have maintained chronology between input parameters in line with storage (i.e., temporal consistency of between input parameters regarding hourly load and RES generation profiles). Authors have also considered different ESSs with different timeframes (i.e., batteries, PSH, Electric Vehicle, Compressed Air Energy Storage, and Concentrated Solar Power). They concluded that ESSs improve the integration of renewables by flattening peak load. However, the case study explored can no longer be considered a highly renewable scenario since it was based on 2020 scenario forecasts and assumed that 20% of the total installed capacity would be renewable. Although operating reserves are considered in the study, the authors do not highlight the challenge that SO will face to ensure the security and continuity of supply of the power system when RES are the largest generation in the mix. As pointed out by BOE (2021) and ENTSO-E (2021b), the frequency of occurrence of critical events should increase; detailed characterization of critical events would provide a rigorous analysis of the complementary and competitive role of the different ESSs. This last aspect is missing in the study presented in Dollinger and Dietrich (2013).

Energy landscapes change with the installation of new nondispatchable RES capacity, and this brings challenges across several dimensions of power systems (i.e., generation, transmission, and distribution). Refs. MITECO (2021, 2020) and BOE (2021) have already assessed the need to increase ESS capacity, mainly with battery and PSH technologies, to balance non-dispatchable RES output. Battery technology, almost mature at the generation level of power systems, lacks operational background. Therefore, comparing the roles of PSH and batteries for optimizing the energy mix is a topic of interest.

When evaluating the impact of energy storage on electricity system operation, the technical characteristics of ESSs need to be modeled. Refs. ENTSO-E (2021b) and ACER (2020) established a list of hypotheses around the methodology to follow to model a hydropower plant. These studies address the aggregation of hydraulic power plants and water reservoirs according to their nature and water inflows' inputs. Both references emphasized the distinction between Closed- and Open-Loop Pumped Storage Hydro (CLPSH and OLPSH). Since CLPSH shows different technical parameters than OLPSH, such as energy reservoir capacity and energy inflows, mixing two sub-technologies can lead to misestimating the role of each of them. In the remainder of this article, ESS refers only to PSH installations (i.e., OLPSH and CLPSH) and batteries, unless otherwise specified.

In Chazarra et al. (2016), the authors presented a hydrothermal operation planning model considering the energy and reserve market. Although their contribution, which was also listed in the review of Gonzalez et al. (2014), lies in modeling the use of operating reserves in a deterministic way based on historical data, their analysis does not account for high non-dispatchable RES penetration. Additionally, the explored scenarios do not consider batteries nor distinguish the subcategories belonging to the PSH (i.e., OLPSH and CLPSH). In Denholm et al. (2020), the authors have analyzed the role of batteries as a peaker capacity in several US regions in scenarios with a high share of non-dispatchable RES. Using a firm capacity approximation based-method similar to the Effective Load Carrying Capability (ELCC) (Madaeni et al., 2013), their research concludes that battery technology has the potential to behave as a peaker capacity, and this role will increase as the share of solar PV increases. Despite the relevant conclusion, the studies presented in Chazarra et al. (2016) and Denholm et al. (2020) lack consistency in comparing the roles of different ESS. Firstly in Denholm et al. (2020), the authors assume a perfect forecast and therefore neglect the operating reserves. Secondly, the technical differences of the different ESSs are not considered either in Chazarra et al. (2016) or Denholm et al. (2020). Thus, according to insights from Refs. Heuberger et al. (2017), REE (2009), Diaz et al. (2014), ENTSO-E (2021b) and ACER (2020), applied methodologies in Chazarra et al. (2016) and Denholm et al. (2020) are not technology-neutral.

Assessing operating roles of different ESS with different timeframes requires considering a time scale long enough to take into account the variability of the RES with a resolution small enough to best accommodate the operating reserves. In addition, the temporal consistency between the data in line with storage timeframes goes along with the study of how hydro storage technologies and batteries can alleviate variability and non-dispatchability of solar and wind resources (ENTSO-E, 2021b; ACER, 2020). Thus, input data must retain the chronological consistency between the variables (solar, wind, water inflows, and load demand) and the model must consider every hour of the year and not representative periods. This feature is particularly relevant since critical events such as high demand, low renewable production, and low hydro inflows, and unavailability of thermal units can range from a minute to several weeks. Their characterization within the model should be sufficiently precise to analyze the role of storage during these periods.

In Werlang et al. (2021), the authors jointly used an expansion and operation optimization model with other simulation modules

to highlight the impacts of regulation on the use of ESSs in two case studies: Brazil and Mexico. One of their conclusion is that the roles of ESSs change depending on the power system considered. According to the authors' conclusions, batteries would have a more critical operational role in Mexico than in Brazil due to the diversity of the energy mix and the flexibility requirements that the power system may have. On the one hand, being energetically dependent on hydropower generation. Brazil presents energy flexibility requirements that batteries cannot meet due to their low energy capacity. On the other hand, Mexico, where the mix is more diversified than in Brazil, shows power flexibilities requirements, leaving battery technology a role to play. Although their study yields significant findings, the methodology used in Werlang et al. (2021) does not follow modeling guidelines brought in ENTSO-E (2021b) and ACER (2020). Indeed, by using representative days, the authors no longer maintain the chronology and consistency between the different time series.

This article proposes a comparison between ESSs with different timeframes that will be part of the future ESS mix, using a medium-term operation planning model. Contributions of this paper include:

- Jointly considering energy and balancing services accounting for the proper distinction between balancing capacity and balancing energy
- A medium-term operation optimization model preserving the chronology and consistency of input data time series
- The proper modeling of different ESS with different storage timeframes and quantification of ESS impact on the operation of power systems
- A detailed hydrothermal operation planning modeling according to market-cleared units for storage hydro and physical individual plants for PSH.

3. Methodology

Analyzing the operational competition between a mix of different ESSs requires an hourly resolution at the operation level and keeping the chronology of events (in terms of demand and non-dispatchable RES production) over a medium-term scope. When it comes to model details of the electricity system operation, linear optimization models are often used because of high algorithm efficiency, ensuring users find an optimal solution (Ventosa et al., 2005).

SEED (Spanish Electricity Economic Dispatch), the model presented in this paper, is an optimization medium-term operation support tool for carrying out different case studies or sensitivity analysis that can help representing the generation system operation in the future, for example, 2030. It reproduces the centralized operation of an electricity system on an hourly basis over a time scope of one year (i.e., 8760 h). The model is based on constraints whose main ones are explicitly included in the paper. The SEED model formulation is based on a publicly available open-access model named open TEPES (Ramos et al., 2022).

The model considers hourly demand with provision for balancing services (i.e., balancing capacity and balancing energy) as in Naversen et al. (2020). Although operating reserves occur over short time horizons, hourly approximation within mediumterm models is acceptable. However, distinguishing between the power availability of the unit and the energy delivered remains preferable. Some references link availability and energy stochastically, and others deterministically. One deterministic method links balancing capacity and energy by an activation coefficient based on historical values (Gonzalez et al., 2014). Besides providing information on the management of the different ESSs, it is possible to measure how each technology contributes to providing hourly operating reserves and energy generation since the model preserves input data chronology.

SEED is a hydrothermal medium-term operation planning model that allows splitting hydroelectric power plants according to ACER (2020) and ENTSO-E (0000b). SEED represents storage hydro, CLPSH, OLPSH, and run-of-river. Run-of-river hydro can be considered non-dispatchable RES since it does not have storage capacity. In addition, the model defines different ESSs regarding their efficiency, installed capacity, and maximum and minimum energy capacity reservoirs. Internally, according to those parameters and the ratio of the installed power on the maximum reservoir capacity, the model decides how and which ESS should generate, store, pump, or spill water.

Another input parameter is defined to constrain the operation of CLPSHs and batteries to distinguish between daily operating storage (Daily ESS) and weekly operating storage (Weekly ESS). Weekly ESS can charge at weekends and discharge at every weekday peak hour. The Daily ESS can charge during off-peak hours and discharge at peak hours. All ESSs with a maximum discharge time of 6 h or less are considered as daily storage. SEED models hydro market units, and OLPSHs as seasonal storage. Maximum time discharge for market-cleared hydro units and OLPSHs is set to one week.

Installed capacity, variable costs, weekly energy inflows, and hourly profile of demand and non-dispatchable generation are introduced as data input according to MITECO (2021), ENTSO-E (0000a), ACER (2020), ENTSO-E (0000b) and ENTSO-E (0000b). Similarly to Refs. Dollinger and Dietrich (2013), Dietrich et al. (2012) and ENTSO-E (0000c), SEED considers the hourly supply of upward and downward operating reserves. Since providing operating reserves ensures electricity system continuity, only dispatchable technologies can participate in, such as thermal, ESS facilities and storage hydro (i.e., dams).

The model formulation is detailed below. All parameters and variables presented are written in upper and lower-case letters, respectively.

The objective function (1) to be minimized is the total system operation cost over the full-time scope (one year). The total system operation cost consists of generation production costs CV_g (i.e., fuel, variable O&M, emissions costs), cost of charging ESSs, CV_s , and non-served energy cost (*CENS*). The parameter DUR_n represents the duration of each load level n considered, in this case, $DUR_n = 1$. The set g refers to all installed generating units, while the set *ess*, relates only to generating units able to consume by charging or pumping.

$$\min\left(\sum_{\omega,n,g} P_{\omega} DUR_n(CV_g gp_{\omega,n,g} + CV_s gc_{\omega,n,s} + CENSens_{\omega,n})\right)$$
(1)

Eq. (2) shows the balance between generation and demand. The equation applies simultaneously for all the considered load levels over the time scope.

$$\sum_{g} (gp_{\omega,n,g} - gc_{\omega,n,s}) + ens_{\omega,n} = D_{\omega,n} \qquad \forall \omega, n \qquad (2)$$

Generation output variable of generation units $gp_{\omega,n,g}$ is defined in (3) and considers both the unit energy generation and the unit energy usage – i.e., energy activation – related to the provision of upward and downward operating reserves. The model considers the energy activation for providing upward and downward operating reserves in a deterministic way as in Dollinger and Dietrich (2013), Chazarra et al. (2016) and Pérez-Díaz et al. (2015). Energy activation (i.e., energy usage $ad_{\omega,n,or}$, $au_{\omega,n,or}$) is modeled using deterministic activation coefficients (*CD*, *CU*) applied to the value of the operating reserve provided ($dr_{\omega,n,or}$,

 $ur_{\omega,n,or}$) as shown in Eq. (19). Eq. (3) does not consider nondispatchable technologies.

$$gp_{\omega,n,g} = \underline{P_g}uc_{\omega,n,g} + p_{\omega,n,g} + au_{\omega,n,or} - ad_{\omega,n,or} \quad \forall \omega, n, g \quad (3)$$

The consumption output variable of installed ESS units $gc_{\omega,n,s}$ is defined in (4). It includes the unit energy consumption and the unit energy activation related to the provision of upward and downward operating reserves. Variables that account for ESS participation in operating reserves $(dr'_{\omega,n,s}, ur'_{\omega,n,s})$ when they are consuming (charging) are different from the ones used in Eq. (3) which refers to them being producing. Since ESSs can generate and pump whereas thermal units do not, the model needs to differentiate the upward and downward operating reserves provided by ESS when pumping (Eq. (3)) than when producing (Eq. (2)). Therefore, the model distinguishes ESS from non-ESS technologies according to the supply of operating reserves.

$$gc_{\omega,n,s} = c_{\omega,n,s} + au'_{\omega,n,s} - ad'_{\omega,n,s} \quad \forall \omega, n, s$$
(4)

The logical relationship between commitment, start-up, and shutdown status of a committed unit is presented in (5) based on Tejada-Arango et al. (2020). Eq. (5) takes into account for the first load level of the first period an initial commitment status for all units, based on merit order loading, including non-dispatchable RES and ESS units. (n - 1) is used for the previous load level.

$$uc_{\omega,n,g} - uc_{\omega,n-1,g} = su_{\omega,n,g} - sd_{\omega,n,g} \quad \forall \omega, n, g$$
(5)

Maximum ramp-up and ramp-down for both thermal and ESS units are respectively formulated in the Eqs. (5a) and (5b) according to Ref. Damci-Kurt et al. (2016). Eqs. (5a) and (5b) reflect the capacity $p_{\omega,n,g}$ and balancing capacity $ur_{\omega,n,g}$ delivered in the hour under consideration, as well as the capacity $p_{\omega,n-1,g}$ and balancing capacity services $ur_{\omega,n-1,g}$ provided in the previous hour.

$$\frac{p_{\omega,n,g} + ur_{\omega,n,g} - (p_{\omega,n-1,g} + dr_{\omega,n-1,g})}{RU_g DUR_n}$$

$$\leq uc_{\omega,n,g} - su_{\omega,n,g} \quad \forall \omega, n, g$$
(5a)

$$\frac{p_{\omega,n,g} - dr_{\omega,n,g} - (p_{\omega,n-1,g} + ur_{\omega,n-1,g})}{RD_g DUR_n}$$
(5b)
$$< -uc_{\omega,n,g} + sd_{\omega,n,g} \qquad \forall \omega, n, g$$

Eq. (6) defines the state of charge of energy reservoir for all ESS units. The model considers roundtrip efficiency EF_g for the consumption $g_{c_{\omega,n,s}}$ of an ESS unit. Although hydropower production output actually changes according to the level of energy capacity reservoirs, the model assumes that the production output of ESS $gp_{\omega,n,s}$ (i.e., storage hydro and PSH) do not have water head dependency. In other words, generating and pumping actions are reflected in the state of charge of different energy reservoirs. At the same time, the available power capacity is assumed to be constant regardless of the level of energy reservoir. That means the power output does not vary according to the state of charge of the energy reservoir. Energy inflows $EI_{\omega,n,s}$ are assumed to be determined weekly. ESS state of charge is formulated according to the duration of the ESS discharge parameter τ_s and therefore limits the use of the ESSs according to their maximum discharge time. For example, in case ESS has $\tau_s = 24$ or 168 Eq. (6) will set the ESS state of charge variable one time per day or per week, respectively, over the year (i.e., 360 times or 52 times in a year). Therefore, different levels of flexibility brought by ESSs with different timeframes according to their discharge time are accounted for. State of charge of energy reservoir for ESS also considers the possible spillages $s_{\omega,n,s}$ and the energy usage from providing operating reserves as set in (3) and (4).

$$SoC_{\omega,n-\tau_{s},s} - SoC_{\omega,n,s} - s_{\omega,n,s} + \sum_{n'=n+1-\tau_{s}}^{n} \times (EI_{\omega,n,s} - gp_{\omega,n,s} + EF_{g}gc_{\omega,n,s}) = 0 \quad \forall \omega, n, s$$
(6)

Upward and downward operating reserves provided by dispatchable units, such as thermal and ESS units, are shown in Eqs. (7) and (8). Considering that ESS can provide auxiliary services while pumping, variables for representing the supply of operating reserves are differentiated for thermal and ESS units when producing $(dr_{\omega,n,or}, ur_{\omega,n,or})$ and ESS units when charging $(dr'_{\omega,n,or}, ur'_{\omega,n,or})$. The restrictions (7) and (8) are such that the operating reserves to supply at each load level must be greater or equal to the input parameters $(DR_{\omega,n}, UR_{\omega,n})$. Thus, the model accounts for operating reserves as an additional load to supply and provides a safety margin in case of critical events (Dollinger and Dietrich, 2013; Dietrich et al., 2012; ENTSO-E, 0000c; Dietrich et al., 2010).

$$\sum_{or} ur_{\omega,n,or} + \sum_{s} ur'_{\omega,n,s} \ge UR_{\omega,n} \quad \forall \omega, n$$
(7)

$$\sum_{or} dr_{\omega,n,or} + \sum_{s} dr'_{\omega,n,s} \ge DR_{\omega,n} \quad \forall \omega, n$$
(8)

Eqs. (9) and (10) below define the maximum and minimum output of the second block of a committed unit. Eqs. (9) and (10) consider providing downward and upward operating reserves $(dr_{\omega,n,or}, ur_{\omega,n,or})$. Eqs. (9) and (10) are formulated for thermal units and ESS units only in the case of generating (i.e., discharging).

$$\frac{p_{\omega,n,or} + ur_{\omega,n,or}}{\overline{P_{or}} - P_{or}} \le uc_{\omega,n,or} \qquad \forall \omega, n, or$$
(9)

$$\frac{p_{\omega,n,or} - dr_{\omega,n,or}}{\overline{P_{or}} - \underline{P_{or}}} \ge 0 \qquad \forall \omega, n, or$$
(10)

Eqs. (11) and (12) define ESS unit consumption by considering supplying upward and downward operating reserves from ESS $(u'_{\omega,n,s}, dr'_{\omega,n,s})$ while the unit is consuming (i.e., charging). Eqs. (9)–(12) were adapted from Tejada-Arango et al. (2020), Gentile et al. (2017), Morales-Espana et al. (2014) and Morales-Espana et al. (2013).

$$\frac{c_{\omega,n,s} + dr'_{\omega,n,s}}{\overline{c}_s} \le 1 \qquad \forall \omega, n, s$$
(11)

$$\frac{c_{\omega,n,s} - ur'_{\omega,n,s}}{\overline{c}_s} \ge 0 \qquad \forall \omega, n, s$$
(12)

Eq. (13) prevents the ESS units from charging and discharging simultaneously using their full installed capacity. As it can be noted, Eq. (13) is consistent with Eqs. (9) and (11).

$$\frac{p_{\omega,n,s} + ur_{\omega,n,s}}{\overline{P_s} - P_s} + \frac{c_{\omega,n,s} + dr'_{\omega,n,s}}{\overline{c}_s} \le 1 \qquad \forall \omega, n, s$$
(13)

Eqs. (14) and (15) prevent ESS units while producing for providing operating reserves (i.e., upward and downward, respectively) if their energy reservoir is empty.

$$ur_{\omega,n,s} \leq \frac{SoC_{\omega,n,s}}{DUR_n} \quad \forall \omega, n, s$$
 (14)

$$dr_{\omega,n,s} \le \frac{I_s - SoC_{\omega,n,s}}{DUR_n} \quad \forall \omega, n, s$$
(15)

Eq. (16) is employed to respect the ratio between the upward and the total reserves provided set by the SO for all units. Although this restriction also applies to ESSs when charging and discharging $(dr'_{\omega,n,s}, ur'_{\omega,n,s})$, for the sake of simplicity, it is not shown here.

$$ur_{\omega,n,g} = R_{\omega,n} \left(\frac{ur_{\omega,n,g} + dr_{\omega,n,g}}{DUR_n} \right) \quad \forall \omega, n, g \in or$$
(16)

Eqs. (17)–(19) handle the relation between balancing capacity and balancing energy provided for generation plants allowed to participate in operating reserves. Eqs. (20)–(22) handle the same relation as (17)–(19) respectively to ESS in pumping mode. Eqs. (17), (18) and (20), (21) include binary variable $\varphi_{\omega,n,or}$ to account for the activation of the balancing capacity provided by thermal and ESS units. Eqs. (19) and (22) prevent units from supplying upward and downward balancing energy simultaneously. Eqs. (17)–(22) are inspired from Naversen et al. (2020) and Pérez-Díaz et al. (2020).

$$au_{\omega,n,or} \leq (P_{or} - \underline{P_{or}}) \varphi_{\omega,n,or} \quad \forall \omega, n, or \quad (17)$$

$$ad_{\omega,n,or} \leq \left(\overline{P_{or}} - \underline{P_{or}}\right) \left(1 - \varphi_{\omega,n,or}\right) \quad \forall \omega, n, or$$
(18)

$$au_{\omega,n,or} - ad_{\omega,n,or} = CU \cdot ur_{\omega,n,or} - CD \cdot dr_{\omega,n,or} \quad \forall \omega, n, or$$
(19)

$$au'_{\omega,n,s} \leq \left(\overline{C_{or}} - \underline{C_{or}}\right)\varphi_{\omega,n,s} \quad \forall \omega, n, s \quad (20)$$

$$ad'_{\omega,n,s} \leq \left(\overline{C_{or}} - \underline{C_{or}}\right) \left(1 - \varphi_{\omega,n,s}\right) \quad \forall \omega, n, s \quad (21)$$

$$au'_{\omega,n,s} - ad'_{\omega,n,s} = CU \cdot dr'_{\omega,n,s} - CD \cdot ur'_{\omega,n,s} \quad \forall \omega, n, s \quad (22)$$

Eqs. (23)–(24) prevent ESS units while charging for providing operating reserves (i.e., upward and downward, respectively) if their energy reservoir is empty.

$$ur'_{\omega,n,s} \le \frac{I_s - SoC_{\omega,n,s}}{DUR_n} \quad \forall \omega, n, s$$
 (23)

$$dr'_{\omega,n,s} \le \frac{I_s - SoC_{\omega,n,s}}{DUR_n} \quad \forall \omega, n, s$$
(24)

Eqs. (25)–(35) detail variable bounds.

$$0 \le gp_{\omega,n,g} \le \overline{P_g} \qquad \forall \omega, n, g \tag{25}$$

$$0 \le gc_{\omega,n,g} \le C_g \qquad \forall \omega, n, g$$

$$0 \le ur_{\omega,n,g} \le \overline{P_g} - P_g \qquad \forall \omega, n, g$$
(26)
(27)

$$0 \le ur'_{\omega,n,s} \le \overline{C_s} \qquad \forall \omega, n, s \qquad (28)$$

$$0 \le dr_{\omega,n,g} \le \overline{P_g} - \underline{P_g} \qquad \forall \omega, n, g \tag{29}$$

$$0 \le u_{\omega,n,s} \le C_s \qquad \forall \omega, n, s \qquad (30)$$
$$0 \le p_{\omega,n,g} \le \overline{P_g} - P_g \qquad \forall \omega, n, g \qquad (31)$$

$$0 \le c_{\omega,n,g} \le \overline{\overline{C}_s} \qquad \forall \omega, n, s \tag{32}$$

$$0 \le s_{\omega,n,s} \qquad \forall \omega, n, s \tag{33}$$

$$0 \le SoC_{\omega,n,s} \le I_s \qquad \forall \omega, n, s \tag{34}$$

$$0 \le ens_{\omega,n} \le D_{\omega,n} \qquad \forall \omega, n \tag{35}$$

The contribution of the model lies in the fact that it is a medium-term model with hourly operation details that simultaneously differentiates between different ESS, according to their time frames, and considers the energy used by ESS and thermal units in providing energy and operating reserves.

4. Case study: The Spanish electricity system in 2030

The Spanish mainland's electricity system will be used as a case study given non-dispatchable RES shares are expected to reach 74% of total electricity consumption by 2030 according to NECP (MITECO, 2021). Based on Spanish NECP 2030, Table 1 summarizes parameters of installed generation capacity per technology in Spain by 2030. This study does not consider the effect of cross-border interconnections.

Representation in SEED of electricity system of the Spanish peninsula in 2030 according to MITECO (2021), Diaz et al. (2014) and Morales-Espana et al. (2014).

Technologies	Installed capacity (# programming units considered)	Installed pump capacity	Energy reservoirs	Cycle discharge	Roundtrip efficiency	Variable cost	Emission rate	O&M Variable cost
	[MW]	[MW]	[GWh]	[Seasonal/Weekly/Daily]	[%]	[€/MWh]	[€/MWh]	[€/MWh]
Nuclear	3050 (3)	na	na	na	na	23	0	0
CCGT	24560 (50)	na	na	na	na	40	0,33	2
Cogeneration	3980 (1)	na	na	na	na	0	0,575	0
Solar PV	38 404 (122)	na	na	na	na	0	0	0
Solar Thermal	7300 (1)	na	na	na	na	0	0	0,46
Wind Onshore	48 550 (70)	na	na	na	na	0	0	0
UGH	7500 (53)	na	9780	Seasonal	na	0	0	0
No UGH (RoR)	1303 (1)	na	na	na	na	0	0	0
OLPSH	7750 (4)	2114	6208	Seasonal	0,75	0	0	0
Existing CLPSH	3648 (10)	3552	120	Weekly/Daily	0,75	0	0	0
PCI I	235	235	1,5	Daily	0,79	0	0	0
PCI II	3400	3400	27,2	Weekly	0,79	0	0	0
PCI III	552	548	3,67	Daily	0,78	0	0	0
Batteries	2500 (1)	2500	10	Daily	0,9	0	0	0
Others RES	1730 (1)	na	na	na	na	0	0	0

According to MITECO (2021, 2020) and BOE (2021), ESSs are identified as necessary in integrating non-dispatchable RES. The available storage will be a mix of ESS technologies responding to different technical and economic characteristics. Therefore, it is essential to consider the features of all ESSs considered to evaluate their operation in the future mix at a programming unit¹ level.

When representing the system's operation to assess the ESS role in high shares' scenario of non-dispatchable RES, it would be preferable to define the ESS mix in the model as suggested in Ref. Ventosa et al. (2005). Therefore, the model can effectively evaluate the flexibility provided by each of the available ESS and their impact on the system's operation when the electricity system faces high shares of non-dispatchable energies (ENTSO-E, 2021a).

According to NECP, in 2030, the Spanish electricity system will dispose of 12 GW of ESS composed of 9.5 GW of PSH and 2.5 GW of batteries. In 2020, Spain counted 5.6 GW of existing PSH installed capacity. Therefore, almost 3.9 GW of CLPSH should be installed up to 2030 (MITECO, 2021; ENTSO-E, 0000a).

Existing installed CLPSH programming units range from 100 MW to almost 1400 MW. Moreover, the maximum water energy reservoir capacity for CLPSH currently installed in Spain varies from approximately from 1 GWh to 31 GWh (REE – activities, 2020). As a result, the maximum consecutive discharge time when the reservoir is at maximum capacity can range from five hours to more than one week. Therefore, to study the complementarity roles and operational competition of ESSs for Spain in the horizon of 2030, energy capacity reservoir and roundtrip efficiency for projected ESSs (i.e., PSH and batteries) remain sensible parameters to be defined.

ESSs such as batteries and PSH are characterized by their installed power, maximum discharge time when the reservoir is full (i.e., or maximum energy reservoir capacity), and roundtrip efficiency such as defined by ENTSO-E (2019). Refs. MITECO (2021) and ENTSO-E (0000a) specify that installed batteries will have a minimum of two hours of discharge with an installed power of 2.5 GW. Given the roundtrip efficiency of batteries is not specified in MITECO (2021) or ENTSO-E (0000a), this parameter is taken from Mongird et al. (2019). Regarding the new CLPSHs projected to be installed, only the total installed power is specified in MITECO (2021) and ENTSO-E (0000a). According to the technical parameters of batteries and new CLPSHs, the following assumptions are formulated:

- The roundtrip efficiency of the batteries is based on Mongird et al. (2019). MITECO (2021) and ENTSO-E (0000a) mention that discharge time at maximum capacity of batteries must be at least 2 h, suggesting that it could be higher. Since various references represent Lithium-Ion battery technology with a discharge time of 4 h, a discharge time of 4 h has been adopted in the model as an average value of maximum discharge time for battery energy storage.
- Technical and economic parameters for new CLPSHs are based on current ongoing projects, named Project of Common Interest (PCI) (PCI, 2021), aimed to be finished and operating by 2030. Table 1 shows three projects of new CLPSH, namely PCI I, PCI II, and PCI III, and their technical features, which are included in the analysis.

Apart from the heterogeneous size of the future ESS mix, Spain has the same installed capacity in Combined Cycle Gas Turbine (CCGT) as storage hydro, i.e., around 16% of its total installed generation mix. According to ten years of historical data (REE – Activities, 2020), annual hydropower production ranged between 13 GWh in 2005 and 40 GWh in 2013. Even though a similar pattern is repeated every year for the hydropower production of the Spanish peninsular system, there is a change in magnitude from year to year, resulting in an annual variability. In addition to a yearly variability, the water inflows data also show seasonal variability. For illustrating hydropower seasonality dependence, Fig. 1 represents the daily productions during February and August for 2015 and 2019 for Spain based on hourly values from ESIOS (2020).

Fig. 1 shows how much the trend for the same month from one year to another can vary (e.g., March 2015 and 2019 have different trends according to Fig. 1). In addition, Fig. 1 allows seeing the changes in the order of magnitude of daily output for the same month in two different years. Therefore, representing hydropower technology in the Spanish electricity system must consider annual and seasonal water inflow variations. The representation of the hydrothermal system operations at an hourly level over one year and having a large historical database to explore the most extreme scenarios is validated by MITECO (2021), ENTSO-E (2021b), ACER (2020) and ENTSO-E (0000c).

Load demand and generation hourly profiles introduced in the model are based on 2015 data (REE – Activities, 2020). Hourly generation profiles of solar PV and wind generation are introduced for each programming unit considered by SEED. As wind and sun are not blowing and shining in the same way at different

¹ Spanish SO defines Programming units as "thermal unit, pumping unit, management unit of hydroelectric power plants or management unit of a set of wind turbines in a wind farm which evacuate energy in the same grid node" (Glossary, 2021).



Fig. 1. Daily hydropower production data over February and August in 2015 and 2019 according to available data (Pérez-Díaz et al., 2020).

places, SEED can maintain differences in production according to programming unit sites. However, for this study, Solar PV and wind turbine programming units were aggregated according to the company they belong to. The same treatment is given to water inflow profiles according to hydro programming unit and OLPSH. Water inflows between two different hydro basins are not totally correlated because of their various sites. (e.g., in the Spanish peninsula, the Ebro basin experiences a late winter snowfall that the Duero basin does not have.)

Thermal power plants are defined by the installed capacity, Equivalent Forced Outage Rate (EFOR) based on historical values, emission rate, and variable costs parameters. Renewable non-dispatchable technologies (Wind and Solar PV) do not have variable costs, and they follow hourly generation profiles.² Hydropower plants are defined by installed capacity (MW), EFOR³ (%), reservoir capacity (MWh), maximum time discharge cycle (hours), and roundtrip efficiency (%).

The hourly operating reserve requirements are based on three parameters according to demand and forecasting errors of wind generation (3% and 10% respectively as in Dietrich et al. (2010)) and the failure of the biggest generation unit.⁴ The energy provided by the committed power reserve capacity is based on historical data (REE – Activities, 2020). In Spain, balancing energy represents 25% and 30% (i.e., *CD* and *CU*) of downward and upward operating reserves to be provided, respectively, according to Chazarra et al. (2016).

Table 1 resumes the main input parameters for representing the Spanish electricity system regarding the NECP objective (MITECO, 2021) based on scenarios presented in ENTSO-E (0000a). Table 1 shows variable costs (i.e., fuel costs), O&M variable cost and emission rate according to IEA (2019) considered, and technical parameters of generating units present in the system studied. Table 1 also presents the required parameters for defining ESS according to MITECO (2021) and ENTSO-E (0000c), such as installed power capacity, energy reservoirs associated with ESS technologies, roundtrip efficiency.

Table 1 shows that the ESS mix can be heterogeneous in terms of maximum energy reservoir capacity, storage duration cycle, and installed capacity. Table 1 details how SEED disaggregates hydropower technologies into four technologies (i.e., programming unit -UGH-, Run of the river,⁵ OLPSH, and CLPSH). Energy reserves capacity (GWh) associated with UGHs and OLPSH are considered similarly as REE - Activities (2020). As mentioned in Oyler and Parsons (2020), the total energy reservoir capacity of 18538 GWh available in Spain does not include the energy reservoir capacity of CLPSH.⁶ Different methods exist to calculate the maximum energy reservoir capacity of CLPSH according to the net waterfall (Latorre et al., 2005). In this paper energy reservoir capacity of each existing CLPSH is based on the maximum consecutive working hours multiplied by the capacity installed.⁷ Pumping efficiency considered for OLSPH and existing CLPSH is set according to Oyler and Parsons (2020), Kougias and Szabó (2017), and for new CLPSH according to PCI (2021). The collection of data in line with hourly generation profiles and the aggregation considered for hydropower programming units have both followed the recommendations of BOE (2020), Ventosa et al. (2005) and ACER (2020) according to temporal consistency between input data and programming units.

Until now, Spain has relied on the flexibility provided by fossil-fuel generation and hydropower technologies. Although the storage capacity will increase, it is important to know how the different ESSs will ensure the system's continuity of supply by increasing its flexibility.

 $^{^2}$ Other sources such as biomass and cogeneration are considered also with generation profiles and are non-dispatchable. Future research should consider the modeling of these resources considering the constraints associated with them.

 $^{^3}$ In the case of hydropower, EFOR is used to consider the capacity limit due to not enough water at the reservoir and it is based on a historical series of hourly production over 10 years. It allows us to consider the capacity reduction due to water-head effect. However, it does not apply to the CLPSH, as the historical data have shown that they have operated at maximum installed capacity.

 $^{^4}$ For upward operating reserves it is based on the failure of a biggest plant and for the downward operating reserves it is based on the failure of the biggest pump physical unit, in Spain 1100 and, 200 MW, respectively according to REE – Activities (2020) and ESIOS UP (2020).

 $^{^{5}}$ Non-UGH includes non-dispatchable hydropower technology, also known as Run-of-River.

 $^{^{6}}$ This value represents the maximum physical value. However, historically the maximum reservoir level reached in Spain was around 14 TWh in 2016 (ESIOS, 2020).

⁷ The energy reservoir capacities are obtained from a 10-year database providing hourly data for each existing CLPSH.

Base case (*BC*) and scenarios sensitive regarding water inflows (*BCdry*, *BCwet*) and the modeling of balancing services (*BC no balancing energy*, *BC no balancing capacity*).

	ВС	BCdry	BCwet	BC no balancing energy	BC no balancing capacity
Energy activation	х	х	х		
Unit power reserve	х	х	х	х	
Water inflows	Medium	Low	High	Medium	Medium

Table 3						
Problem size of BC scenario.						
Constraints	16.690.570					
Variables	18.093.718					
Non-zeros	56.888.301					

What is more, given the high variability introduced by nondispatchable RESs, critical events, where ESSs are expected to play a key role, are likely to increase. Hence, critical events must be tracked and analyzed. Analyzing the role of storage in Spain in 2030 through an operation model must be done with the following considerations:

- Medium-term scope to consider hydropower seasonal variability over a year.
- Maintaining chronology of events for analyzing ESS role during critical events.
- Representing different ESSs with different timeframes and technical parameters to characterize rigorously ESS roles in future Spain 2030 mix.
- Since electricity system security became challenging due to the penetration of non-dispatchable RESs, it is necessary to consider balancing services through operating reserves' modeling is necessary. Also, given hydropower technologies strongly participate in providing operating reserves, balancing services must taken into account when evaluating the role of ESS.

5. Scenarios

Exploring the role of ESS in a 2030 scenario according to MITECO (2021) requires firstly building the base case (BC), which will be used as a reference case as represented in Table 1. A scenario of average water inflows – i.e., 25.1 TWh according to the year 2015 (ACER, 2020) – is used as a reference case. Two additional scenarios are defined and compared to the *BC* scenario according to water inflows. *BCwet* and *BCdry* are based on the annual weekly profile of 2016 and 2017, respectively, with 34.5 TWh and 15.9 TWh of energy hydro inflows.

In addition to scenarios exploring water inflows' influence on ESS operation, two more scenarios are designed to contrast the *BC* results based on different modeling assumptions of balancing services. Although modeling balancing services is approximated by hourly upward and downward operating reserves, the distinction between unit power reserve and energy activation is maintained. Therefore, balancing energy is not considered in the first scenario based on *BC (BC no balancing energy)*. To complete exploring the consequences of including balancing services in a medium-term model, this study analyzes another scenario based on *BC without considering the provision of balancing capacity*. Table 2 resumes the five scenarios explored according to both parameter and constraint changes.

All scenarios were executed using GUROBI 9.1 under GAMS 35.0 with 32 GB of RAM. Table 3 presents the problem size:

6. Results and analysis

Given the possible operational roles of ESSs, such as PSHs and batteries, their operation may impact various indicators of system operation. Therefore, according to the focus of the study, this evaluation considers the impact on (1) changes in technologies generation and total operation cost according to different water inflows' scenarios, (2) the same changes according to different modeling assumptions of balancing services, (3) sensitivities analysis on ESS parameters, (4) technologies providing operating reserves, (4) behavior of ESSs during critical events.

6.1. Changes in technologies generation and total operation cost according to different water inflows' scenarios

In this section, the roles of ESS technologies are assessed according to different water inflows scenarios. Fig. 2 compares annual energy production per technology for two different years, 2019 according to available data in ESIOS (2020), and 2030 according to results of the BC scenario. As foresees by MITECO (2021), in 2030, non-dispatchable RESs represent most of the electricity generation. Given the energy produced, one might think that CCGT and ESS have different behaviors. However, comparing these technologies in terms of hours of use is necessary as their installed capacity is different.

Figs. 3 and 4 represent the hourly operation during one day in 2019 according to available data from the SO (ESIOS, 2020) and 2030 according to the *BC* scenario, respectively. Note that figures represent the same day (i.e., the first day of the last week of January). Figs. 3 and 4 highlight operating changes in the economic dispatch of different technologies and changes in load patterns.

Comparing 2019 and 2030 economic dispatch allows one to observe the decreasing available baseload in the Spanish electricity system in 2030 (i.e., nuclear technology) by half, and no coal production is present. In addition, CCGT sees its output reduced and its scheduling changed. Moreover, the increase of the power requirement at peak hours is higher in 2030 than in 2019. Indeed, due to the non-dispatchable RES variability, the system requires more flexibility provided by thermal hydropower and ESS technologies.

Additionally, Tables 4 and 5 present results for the *BC*, *BCdry*, *BCwet* scenarios through annual full-time equivalent operation hours⁸ and various electricity system indicators, respectively (i.e., total operation cost, total emission, RES curtailment).

The variations in total operation cost are mainly due to a change in CCGT generation since hydropower and ESS technologies have no variable cost. UGH, and OLPSH productions follow the annual water inflows scenario trend. Although it may seem counter-intuitive, batteries, existing and new CLPSH, increase

⁸ Annual full-time equivalent operation hours is commonly employed to point out and compare the use of different technology. It is defined for each technology by the ratio between the total energy produced and the installed capacity.



Fig. 2. Mix of electricity generation in GWh in Spain in 2019 according to Gentile et al. (2017) and 2030 according to BC scenario results obtained with SEED.



Fig. 3. Hourly economic dispatch for the Spanish electricity system during the day with the annual peak hour in 2019 according to Gentile et al. (2017).

their output in a wet scenario and outperform the CCGT in terms of annual working hours according to Table 4. When water inflows are abundant, the system prefers to rely on ESS rather than CCGT since for the former, the minimum technical output depends on the energy stored and do not have fuel cost. Furthermore, in extreme scenarios (i.e., wet and dry scenarios), the duration and frequency of the critical period change according to the based-case scenario. Even though the available mix is not dispatched in the same way in the different scenarios, Table 5 shows that the NECP RES goal of 74% is achieved for all scenarios, and no Energy Not Served (ENS) is observed.

According to Table 4, in a context of high non-dispatchable RES, CCGT, CLPSH (new and existing), and battery technologies work approximately the same duration of time in one year - e.g., below 1500 h. However, annual full-time equivalent operation hours may not be a sufficient indicator to affirm both technologies are used as peak capacity.

Table 4

Annual full-time equivalent operation hours for technologies according to different water inflows scenarios (BC, BCdry, and BCwet).

	BCdry	BCwet	BC	BCdry/BC	BCwet/BC
	[h]	[h]	[h]	[%]	[%]
Battery	1125	1115	1093	3%	2%
CCGT	1465	1041	1267	14%	-22%
CLPSH	1264	1296	1219	4%	6%
Cogeneration	4623	4623	4623	0%	0%
New CLPSH	1441	1439	1395	3%	3%
Nuclear	7224	7224	7224	0%	0%
OLPSH	1908	3395	2476	-30%	27%
Other RES	6987	6987	6987	0%	0%
No-UGH	4437	3039	4091	8%	-35%
Solar PV	1629	1585	1612	1%	-2%
Solar Thermal	2696	2696	2696	0%	0%
UGH	1641	3242	2381	-45%	27%
Wind Onshore	2285	2226	2257	1%	-1%



Fig. 4. Hourly economic dispatch for the Spanish electricity system during the day with the annual peak hour in 2030 according to the BC scenario. Results were obtained with SEED.

Table :	5
---------	---

Indicators of electricity system operation for three different water inflows scenarios (BC, BCdry, BCwet).

		BCdry	BCwet	BC	BCdry/BC	BCwet/BC
Total operation cost	M€	3.862	3.166	3.536	8%	-12%
Emission	MtCO2	22,5	19,0	20,9	7%	-10%
RES curtailment	%	17%	22%	19%	-12%	12%
%RES	%	75%	78%	77%	-2%	2%
SRMC	€/MWh	45	39	44	3%	-11%
ENS	MWh	0	0	0	0%	0%

6.2. Changes in technologies generation and total operation cost according to different operating reserves' modeling considerations

The three scenarios presented in Table 6 may highlight why the medium-term operation planning model should consider balancing services when analyzing the future role of ESS in a context of high renewable penetration. Until now, medium-term operation planning models did not necessarily represent the correct use of the ESS mainly due to its economic use. Given the high shares of renewables in 2030, the medium-term models are led to consider more details at the operational level, such as balancing services, to allow adequate decision-making on the future operation of the electricity system (Pérez-Díaz et al., 2020).

Tables 6 and 7 presents results for *BC*, *BC* no balancing energy, *BC* no balancing capacity scenarios through annual full-time equivalent operation hours, and various electricity system indicators, respectively (i.e., total operation cost, total emission, RES curtailment).

According to Table 6, although not considering energy activation against not balancing services (i.e. neither balancing capacity nor balancing energy) has a small impact, it results in underestimating the contribution of batteries to provide services by 6% (energy and balancing services). Additionally, Table 7 shows that not considering balancing services underestimates annual total operating cost and RES curtailment.

Table 6 shows that battery technology is the most impacted by a change in accounting for balancing services. Considering balancing services increase full-time equivalent operation hours of batteries and CLPSH. Section 6.3 explains more in detail why the operating reserves consideration impacts more significantly batteries. 6.3. Sensitivity analysis for determining how relevant ESS factors impact electricity system dispatch

This section aims to analyze how modeling operating reserves most impact different ESSs. Two additional scenarios, *BC weekly cycle* and *BC roundtrip* are built to isolate parameters that describe batteries. *BC weekly cycle* differs from *BC* according to the cycle discharge of batteries. While in the *BC* scenario, battery technology must be filled up at the end of the day, in the *BC weekly cycle*, batteries must meet their maximum reserve at the end of the week. *BC roundtrip* differs from *BC* according to the roundtrip efficiency of battery technology. In the *BC roundtrip*, the roundtrip efficiency of batteries is similar to the one of the new CLPSHs (i.e., 79%). Table 8 resumes sensitivy scenarios. Table 9 presents results for sensitivity scenarios according to the annual full-time equivalent operation hours for technologies.

According to Table 9, switching from a daily to a weekly discharge cycle gives the batteries greater flexibility. This gain in flexibility increases the dispatch of the battery technology (i.e., + 39% respectively to the *BC* scenario). The technologies seeing their dispatch decreased are first the new CLPSH (i.e., -12%), then the existing CLPSH (i.e., -8%), and finally CCGT (i.e., -2%). Table 9 shows that the variation in the dispatch of the battery technology is higher for the *BC weekly cycle* than for the *BC roundtrip*, respectively, to the *BC* scenario.

To complete sensitivity analysis, Table 10 shows various electricity system indicators, respectively (i.e., total operation cost, total emission, RES curtailment) that are in line with results shown in Table 9. The change from a daily to a weekly discharge cycle of the batteries allows the system to reduce its total operating costs by 1.2%, according to Table 10. Therefore, a greater

Annual full-time equivalent	operation hours for	or technologies	according to	different	scenarios	based on	the m	nodeling
of operating reserves (BC, E	3C no balancing ca	pacity, and BC	no balancing	energy).				

Technologies	BC [h]	BC no balancing energy [h]	BC no balancing capacity. [h]	BC no balancing energy/BC [%]	BC no balancing capacity/BC [%]
Battery	1093	1033	1028	-6%	-6%
CCGT	1267	1266	1266	0%	0%
CLPSH	1219	1203	1205	-1%	-1%
New CLPSH	1395	1392	1392	0%	0%
OLPSH	2476	2536	2555	2%	3%
UGH	2381	2381	2379	0%	0%

Table 7

Indicators of electricity system operation for three scenarios (BC, BC no balancing capacity, BC no balancing energy).

		BC	BC no balancing energy	BC no balancing capacity	BC no balancing energy/BC [%]	BC no balancing capacity/BC [%]
Total operation cost	M€	3.536	3.533,2	3.533,3	-0,07%	-0,07%
Emission	MtCO2	20,9	20,8	20,8	-0,04%	-0,04%
RES curtailment	%	19,2%	19%	18,9%	-0,88%	-1,16%
%RES	%	76,7%	76,8%	76,8%	0,09%	0,11%
SRMC	€/MWh	44	44	44	0%	0%
ENS	MWh	0	0	0	0%	0%

Table 8

Sensitivity scenarios according to roundtrip efficiency (BC roundtrip) and cycle discharge of batteries (BC weekly cycle).

	BC	BC weekly cycle	BC roundtrip
Roundtrip efficiency	90%	90%	79%
Cycle discharge	Daily	Weekly	Daily

flexibility of the batteries helps to reduce the use of CCGTs and the RES curtailed (i.e., -3.3%).

From these sensitivities (i.e., *BC weekly cycle* and *BC roundtrip*), it is possible to classify the ESS parameters that have the greatest impact on their dispatch and thus explain why battery technology is the most impacted by the consideration of operating reserves. Although closely related to the energy capacity of the reservoir, the longer the discharge cycle (i.e., seasonal), the more flexibility it gives to the technology. As batteries have a daily discharge cycle, they have reduced flexibility compared to CLPSH (i.e., weekly discharge cycle). Additionally, at equal roundtrip efficiency, the new CLPSH is the technology that absorbs the lack of capacity of batteries. Thus, the batteries would increase the availability of remaining ESS during critical events, participating vigorously in providing operating reserves, first because of their daily discharge cycle and then because of their high roundtrip efficiency.

6.4. Technologies providing energy and balancing services

Section 6.1 shows that CCGT, batteries, and CLPSH technologies work few hours per year. However, results provided in Section 6.2 show that those technologies are impacted differently according to different balancing services modeling assumptions. Therefore, this section focuses on how much technologies produce in both energy and balancing services.

Fig. 5 presents the results according to technology share in providing balancing capacity services. UGH and OLPSH represent 45% (24% and 22% respectively) of total balancing services provided, which is in line with REE (2009). Although other technologies participating in reserve markets provide less balancing services, it remains relevant to highlight CLPSH and batteries, and CCGT provide together an equivalent quantity of total balancing services than UGH and OLPSH (13%, 11%, and 16%, respectively).

Additionally, when assessing the roles of different ESS, a relevant indicator is to compare the total balancing services and the total energy provided by each technology. Fig. 6 shows the ratio between both services. If the ratio is close to 1, the technology produces same quantity in the energy market as in the reserve market. If the ratio tends to 0, the technology does not provide balancing services. Results according to the *BC* scenario are shown in Fig. 6.

Regarding results provided in this section, they show that technologies provide their services differently. Regarding Fig. 6, battery technology produces almost the same energy in the energy market as the balancing energy, making it sensitive to which modeling assumptions are applied on balancing services. This result reinforces the idea that analyzing the future role of ESS should be done, including both reserve and energy markets and ESS participation in both markets.

6.5. The behavior of ESS during critical events

An additional aspect to compare the role and operational competitiveness of the different available ESSs technologies is their contribution to the firm capacity required in the system to guarantee the supply. Capacity markets are designed to remunerate this very important service. Therefore, this service represents a relevant issue when comparing the competitiveness and role of generation technologies and ESSs. Although this should be properly addressed by using stochastic models, deterministic operation models such as the one described in this paper could be used to get insights into the comparative role of ESSs regarding this issue.

This issue is even more important in systems largely dominated by intermittent generation. A much higher frequency of critical events is to be expected, stressing the system and critical events of much larger duration that should be faced with the non-dispatchable resources in place. Thermal generation and ESSs will be the major players in this regard. In electricity systems dominated by thermal generation, critical hours are strongly related to the number of hours of very high demand. There may be sequences of many consecutive critical hours in electricity systems with a larger share of non-dispatchable RES generation. Analyzing the behavior of ESS technologies during critical events could help in assessing their roles. Therefore, this section evaluates the contribution of PSH units during the most critical hours regarding their production level.

Although several methods exist regarding for assessing peaking capacity resources (Denholm et al., 2020), they are not applicable in deterministic analysis using an operation model without

Annual full-time equivalent operation hours for technologies according to different roundtrip efficiency and cycle discharge of batteries (BC, BC weekly cycle, and BC roundtrip).

Technologies	BC [h]	BC weekly cycle [h]	BC roundtrip [h]	BC weekly cycle/BC [%]	BC roundtrip/BC [%]
Battery	1093	1782	907	39%	-21%
CCGT	1267	1241	1274	-2%	1%
CLPSH	1219	1125	1209	-8%	-1%
New CLPSH	1395	1247	1456	-12%	4%
OLPSH	2476	2459	2433	-1%	-2%
UGH	2381	2381	2351	0%	-1%

Table 10	
----------	--

Indicators of electricity system operation for three scenarios (BC, BC weekly cycle, BC roundtrip).

		BC	BC weekly cycle	BC roundtrip	BC weekly cycle/BC	BC roundtrip/BC
Total operation cost	M€	3.536	3.494	3.548	-1,2%	0,3%
Emission	MtCO2	20,9	20,6	20,9	-1,1%	0,3%
RES curtailment	%	19,2%	18,6%	18,7%	-3,3%	-2,7%
%RES	%	76,7%	76,8%	76,8%	0,4%	0,1%
SRMC	€/MWh	44,3	45	44,3	1,5%	0,1%
ENS	MWh	0	0	0	0%	0%



Fig. 5. Share in % of technologies competing in balancing services according to BC scenario.

considering investments. A different approach is applied in this analysis, where the average value of the capacity factor of the technology among critical periods is used to evaluate the peaking capacity of ESS technologies, that is their contribution to the firmness of the system (Madaeni et al., 2013). The larger they are present in those critical periods, the larger is their ability to contribute to the guarantee of supply of the system.

Regarding the obtained measure, the technology is considered reliable during peak hours of net demand when the measure tends to one. However, the number of critical hours to consider remains to be determined. Indeed, a high number of critical hours would make the measure to tend towards the average annual load factor for each technology since we are including in the analysis hours that are not so critical. On the other hand, a small number of hours would not reflect critical events longer than two hours. It will not capture the risk of not having enough energy during longer close periods, resulting in some ESS being either over or under-evaluated in this regard. Therefore, a sensitivity analysis has been performed concerning the number of critical hours considered, adopting a range between 100 and 300 h. The hours selected are those with the highest net demand once the RES intermittent production is discounted.

Fig. 7 shows capacity factors of ESS technologies based on annual demand (green line), on 100 h and 300 h of highest

demand (blue bars), and on 100 h and 300 h of highest net demand (orange bars).

In Fig. 7, the values of the demand-based capacity factors (blue bars) tend to be close to the annual mean of capacity factor, especially for the critical 300 h rank. Furthermore, the net demand-based capacity factor (orange bar) is higher than the demand-based capacity factor, indicating a better availability of ESS when demand is high and non-dispatchable RES production is low. From these two observations, (1) the so-called critical hours of the system should be defined considering both the production of non-dispatchable RES and load demand; (2) a too large time window could lead to a misinterpretation of the results, as we would no longer talk about availability in critical periods of the considered technology but about the annual mean of the capacity factor.

Another observation in Fig. 7 is the possible relationship between the availability of ESSs and the capacity of their energy reservoirs. Indeed, the obtained ranking of ESS availability at peak hours is in line with the energy reservoir size of each ESS. Regarding results in Fig. 7, the larger the energy reservoir, the more available ESS is.

Changing the critical hours rank affects the availability of ESS technologies distinctly. OLPSH and existing CLPSH see their availability increased when the considered critical hours decrease, whereas the availability of new CLPSH and batteries diminish. As



Fig. 6. Ratio between balancing energy and wholesale energy provided for each technology for BC scenario.



Fig. 7. Capacity factor of ESS technologies obtained for BC scenarios for a range of 100 (left-hand side) and 300 (right-hand side) critical hours according to demand (blue bars) and net demand (red bars). (For interpretation of the references to color in this figure legend, the reader is referred to the web version of this article.)

the observation range of critical hours is reduced, the net demand values are higher. In addition, batteries and new CLPSHs have a smaller energy storage capacity than the remaining ESS (OLPSH and existing CLPSH). Thus, the smaller the number of critical hours considered, the less ESS with small energy reservoirs are available.

Although batteries have a high roundtrip efficiency coefficient, Fig. 8 shows that, among all ESS technologies, batteries would not be available during critical events considering 100 h nor 300 h highest net demand values. This effect can be due because the batteries have a maximum discharge time of 4 h.

Fig. 8 shows that the most extended critical period measured is 7 h. In the case of the highest 100 h of net demand, the number of times 4 h and 5 h events occur is equivalent. In the highest 300 h of net demand, the occurrence of 5 h events is greater than that of 4 h. Although the periods may be of the same duration as the maximum battery discharge time, several periods are ranging from one to three critical hours in a row. In addition, the critical events appear before or after critical periods (e.g., even if they are not consecutive, they are close in time, for instance, during consecutive days). Therefore, due to their low storage capacity, batteries cannot make themselves available at the same level as the remaining ESS (e.g., not having had time to recharge to 100% before the next event).

The data used in this study are exclusive to the case study analyzed. Furthermore, the observations made from the results obtained may be subject to change depending on the case study considered.

7. Conclusions

This article presents a medium-term centralized operation planning model to reproduce the operation of the Spanish system in 2030. However, the proposed approach could be used for a



Fig. 8. Frequency of occurrence of critical events of different durations according to a range of 100 and 300 h of highest net demand for BC scenario.

case study involving a larger geographical area, including France, Spain, and Portugal. The medium-term modeling, including the hourly production of energy and operating reserves, highlighted the role of ESSs in ensuring the security and continuity of supply of the electricity system. Three groups of indicators (e.g., sequences of consecutive hours of highest net demand, total energy and operating reserves production, an average of capacity factor according to a range of peak hours of net demand) were used to assess the contribution of ESSs to the reliability of the electricity system during critical events and to analyze the competition and complementarity between different ESSs (namely, batteries and pumped storage hydro).

According to the ESS availability during critical events, this study highlights that defining critical events should consider demand and non-dispatchable RES production. In addition, the range of peak hours of net demand is a sensitive parameter. It should be chosen cautiously to allow a fair analysis between ESS and the characterization of critical periods of different durations.

In addition, this paper explores the roles of different ESSs according to critical water inflow scenarios (i.e., dry and wet water hydro inflow scenarios). Although RES penetration relies on water hydro inflow scenario considered, impacts on CLPSHs and batteries productions are similar.

Moreover, it has been shown that considering balancing services in a medium-term operation model is relevant to evaluate storage technologies' role. Given that 4 h-batteries provide almost the same amount of energy in both energy and reserve markets, the analysis of their contribution to the electricity system's reliability is sensitive to considering balancing services. Although batteries seem to last to contribute to the system's reliability during critical periods, their production in the reserve market is equivalent to that injected from other ESSs. Therefore, the omission of balancing services might lead to misevaluate analyzing complementarity of different ESSs.

Regarding the generation of ESSs to the energy market, the higher the installed capacity, the higher the production in the reserve market. Besides, the complementarity of the ESSs is that the batteries would essentially be intended to participate in the reserve market and thus allow the larger capacity ESSs such as the CLPSHs to be available to contribute to maintaining the reliability of the system during the highest net-demand peak hours. These conclusions remain in the context of Spain in 2030 according to the planned electricity generation mix and were obtained through deterministic scenarios.

The authors found relevant enhancing operation details modeling within a medium-term operation model for completing this analysis. This change could lead to improving the representation of balancing services. The option of introducing stochasticity is also considered the next step to enhance the study of ESS availability during critical periods. An additional major improvement that could be envisaged is the consideration of demand response and interconnections in the model, as these features, if expected to be large enough, could significantly impact the results of the operational behavior of ESSs in the system. Since the proposed approach only considers the operation of power systems, the first limitation is the expansion of the generation. The proposed approach does not include investment options. However, these could be included in future studies that focus on exploring the impacts of ESS operations on investment decisions and the evolution of the installed energy mix (Ramos et al., 2022). A second limitation relates to the consideration of the transmission network. However, it is not uncommon for the network to be ignored, and even less so for the Spanish case, where, when the study focuses exclusively on generation, the Iberian Peninsula is treated as a single node (i.e., generally referred to as the copper plate) (Morales-Espana et al., 2014; Barquin et al., 2004). A third limitation concerns the stochasticity of the scenarios analyzed. Although the model explores deterministic scenarios, it would be possible to integrate it into a Monte Carlo simulation. Monte Carlo simulation would aim to reproduce many times the different scenarios for demand or RES generation profiles. A fourth limitation is the technologies considered. Although the article proposes to compare different storage systems, some technologies, which are sometimes seen as storage systems, or at least as flexibility solutions, are not included, such as electric vehicles, interconnections, and demand-side management. It is conceivable to introduce these technologies in a future article where the comparison would extend to storage systems and all flexibility solutions available to the system operator. Although, several storage technologies (e.g., Flywheel, Compressed Air Energy Storage, Electric vehicles) are not considered in this study, it would be possible to represent them as long as their parameters (i.e., storage capacity, installed power, efficiency, and discharge time) can be used to model the annual operation. Thus, the proposed approach is scalable and replicable once the parameters are carefully determined.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgments

This research has been carried out thanks to the Spanish Ministry of Economy and Competitiveness MINECO through BC3 María de Maeztu excellence accreditation MDM-2017-0714 Maria de Maeztu Grant. The research has also benefited from the funding of the RETOS COLABORACIÓN program of the Spanish Ministry of Science and Innovation and the Spanish State Research Agency (project "Platform of innovative models to accelerate the energy decarbonization of the economy (MODESC)", with reference number RTC2019- 007315-3). The authors would also like to acknowledge the Iberdrola Chair of Energy and Innovation for their helpful comments.

References

- ACER, 2020. Methodology for the European resource adequacy assessment. [Online]. Available: https://acer.europa.eu/Media/News/Pages/ACER-sets-themethodologies-to-assess-electricity-resource-adequacy-in-the-EU.aspx.
- Barquin, J., Centeno, E., Reneses, J., 2004. Medium-term generation programming in competitive environments: a new optimisation approach for market equilibrium computing. IEE Proc. Gener. Transm. Distrib. 151 (1), 119. http: //dx.doi.org/10.1049/ip-gtd:20040055.
- 2020. Boe-a-2020-16964. 120122, 120317 (196 pages). [Online]. Available: https://www.boe.es/eli/es/res/2020/12/10/(7).
- BOE, 2021. Proyecto de ley de cambio climático y transición energética. Accessed: Jun. 08, 2021. [Online]. Available: https://www.congreso.es/public_oficiales/ L14/CONG/BOCG/A/BOCG-14-A-19-5.PDF.
- Chazarra, M., Garcia-Gonzalez, J., Perez-Diaz, J.I., Arteseros, M., 2016. Stochastic optimization model for the weekly scheduling of a hydropower system in day-ahead and secondary regulation reserve markets. Electr. Power Syst. Res. 130, 67–77. http://dx.doi.org/10.1016/j.epsr.2015.08.014.
- Damci-Kurt, P., Küçükyavuz, S., Rajan, D., Atamtürk, A., 2016. A polyhedral study of production ramping. Math. Program. 158 (1–2), 175–205. http: //dx.doi.org/10.1007/s10107-015-0919-9, Jul.
- Denholm, P., Nunemaker, J., Gagnon, P., Cole, W., 2020. The potential for battery energy storage to provide peaking capacity in the United States. Renew. Energy 151, 1269–1277. http://dx.doi.org/10.1016/j.renene.2019.11.117.
- Diaz, C.A., Gonzalez, P., Campos, Fco. A., Villar, J., 2014. Spanish secondary reserve requirements, clearing and usage. In: 11th International Conference on the European Energy Market (EEM14), Krakow, Poland. pp. 1–5. http: //dx.doi.org/10.1109/EEM.2014.6861266.
- Dietrich, K., Latorre, J.M., Olmos, L., Ramos, A., 2010. Adequate regulation reserve levels in systems with large wind integration using Demand Response. [Online]. Available: https://www.researchgate.net/profile/Kristin-Dietrich-2/publication/228601364_Adequate_Regulation_Reserve_Levels_ In_Systems_With_Large_Wind_Integration_Using_Demand_Response/ links/0a85e536a05afc15fe000000/Adequate-Regulation-Reserve-Levels-In-Systems-With-Large-Wind-Integration-Using-Demand-Response.pdf.
- Dietrich, K., Latorre, J.M., Olmos, L., Ramos, A., 2012. Demand response in an isolated system with high wind integration. IEEE Trans. Power Syst. 27 (1), 20–29. http://dx.doi.org/10.1109/TPWRS.2011.2159252.
- Dollinger, B., Dietrich, K., 2013. Storage systems for integrating wind and solar energy in Spain. In: 2013 International Conference on Renewable Energy Research and Applications (ICRERA), Madrid, Spain. pp. 361–365. http://dx. doi.org/10.1109/ICRERA.2013.6749781.
- ENTSO-E, TYNDP 2018 Final Scenario Report ANNEX I Country Level_Results, ENTSO-E. [Online]. Available: https://eepublicdownloads.entsoe.eu/cleandocuments/tyndp-documents/TYNDP2018/Scenario_Report_Annex_1.pdf.
- ENTSO-E, Hydropower modelling New database complementing PECD. ENTSO-E. [Online]. Available: https://eepublicdownloads.blob.core.windows. net/public-cdn-container/clean-documents/sdc-documents/MAF/2019/ Hydropower_Modelling_New_database_and_methodology.pdf.
- ENTSO-E, TYNDP 2018 Scenario Report ANNEX II Methodology, ENTSO-E. [Online]. Available: https://eepublicdownloads.entsoe.eu/clean-documents/ tyndp-documents/TYNDP2018/Scenario_Report_ANNEX_II_Methodology.pdf.

- ENTSO-E, 2019. Guidance for transmission and storage projects for applying to the TYNDP 2020. [Online]. Available: https://eepublicdownloads.blob. core.windows.net/public-cdn-container/clean-documents/tyndp-documents/ TYNDP2020/190918_ENTSO-E%20Guidance%20for%20transmission%20and% 20storage%20projects%20for%20applying%20to%20the%20TYNDP%202020_ FINAL.pdf.
- ENTSO-E, 2021a. TYNDP 2022 scenario building guidelines. [Online]. Available: https://2022.entsos-tyndp-scenarios.eu/wp-content/uploads/2021/09/2021-10-TYNDP_2022_Scenario_Building_Guidelines.pdf.

ENTSO-E, 2021b. Mid-term adequacy forecast 2020 appendix II. p. 32.

- ESIOS, 2020. Bienvenido ESIOS electricidad · datos · transparencia. https: //www.esios.ree.es/es (accessed May 29, 2020).
- ESIOS UP, 2020. Unidades de programación | ESIOS electricidad · datos · transparencia. https://www.esios.ree.es/es/unidades-de-programacion (accessed May 29, 2020).
- Gentile, C., Morales-Espana, G., Ramos, A., 2017. A tight MIP formulation of the unit commitment problem with start-up and shut-down constraints. Euro J. Comput. Optim. 5 (1–2 SI), 177–201. http://dx.doi.org/10.1007/s13675-016-0066-y.
- 2021. Glossary | red eléctrica de españa. https://www.ree.es/en/glossary (accessed Jan. 17, 2021).
- Gonzalez, P., Villar, J., Diaz, C.A., Alberto Campos, F., 2014. Joint energy and reserve markets: Current implementations and modeling trends. Electr. Power Syst. Res. 109, 101–111. http://dx.doi.org/10.1016/j.epsr.2013.12.013.
- Heuberger, C.F., Staffell, I., Shah, N., Dowell, N.M., 2017. A systems approach to quantifying the value of power generation and energy storage technologies in future electricity networks. Comput. Chem. Eng. 107, 247–256. http: //dx.doi.org/10.1016/j.compchemeng.2017.05.012.

IEA, 2019. World Energy Outlook. IEA, [Online]. Available; https://iea.blob.core. windows.net/assets/98909c1b-aabc-4797-9926-35307b418cdb/WEO2019free.pdf.

- Kougias, I., Szabó, S., 2017. Pumped hydroelectric storage utilization assessment: Forerunner of renewable energy integration or Trojan horse? Energy 140, 318–329. http://dx.doi.org/10.1016/j.energy.2017.08.106.
- Latorre, J.M., Cerisola, S., Ramos, A., Perea, A., Bellido, R., 2005. Simulación de cuencas hidráulicas mediante Programación Orientada a Objetos. [Online]. Available: https://pascua.iit.comillas.edu/aramos/presentaciones/Simulacion% 20de%20cuencas%20hidraulicas%20mediante%20Programacion%20Orientada% 20a%20Objetos.pdf.
- Madaeni, S.H., Sioshansi, R., Denholm, P., 2013. Comparing capacity value estimation techniques for photovoltaic solar power. IEEE J. Photovolt. 3 (1), 407–415. http://dx.doi.org/10.1109/JPHOTOV.2012.2217114.
- MITECO, 2020. Borrador de la estrategia de almacenamiento energético. Accessed: Jun. 08, 2021. [Online]. Available: https://energia.gob.es/es-es/ Participacion/Paginas/DetalleParticipacionPublica.aspx?k=347.
- MITECO, 2021. Borrador actualizado del plan nacional integrado de energía y clima 2021–2030. Accessed: Jun. 08, 2021. [Online]. Available: https://www.miteco.gob.es/images/es/pnieccompleto_tcm30-508410.pdf.
- Mongird, K., et al., 2019. Energy Storage Technology and Cost Characterization Report. PNNL-28866, 1573487, http://dx.doi.org/10.2172/1573487.
- Morales-Espana, G., Latorre, J.M., Ramos, A., 2013. Tight and compact MILP formulation for the thermal unit commitment problem. IEEE Trans. Power Syst. 28 (4), 4897–4908. http://dx.doi.org/10.1109/TPWRS.2013.2251373.
- Morales-Espana, G., Ramos, A., Garcia-Gonzalez, J., 2014. An MIP formulation for joint market-clearing of energy and reserves based on ramp scheduling. IEEE Trans. Power Syst. 29 (1), 476–488. http://dx.doi.org/10.1109/TPWRS. 2013.2259601.
- Naversen, C.Ø., Farahmand, H., Helseth, A., 2020. Accounting for reserve capacity activation when scheduling a hydropower dominated system. Int. J. Electr. Power Energy Syst. 119, 105864. http://dx.doi.org/10.1016/j.ijepes. 2020.105864.
- Oyler, Anthony Fratto, Parsons, John E., 2020. The Value of Pumped Hydro Storage for Deep Decarbonization of the Spanish Grid. MIT Working Papers. [Online]. Available: http://ceepr.mit.edu/files/papers/2020-007.pdf.
- 2021. PCI interactive map. https://ec.europa.eu/energy/infrastructure/ transparency_platform/map-viewer/main.html (accessed Jun. 01, 2021).
- Pérez-Díaz, J.I., Chazarra, M., García-González, J., Cavazzini, G., Stoppato, A., 2015. Trends and challenges in the operation of pumped-storage hydropower plants. Renew. Sustain. Energy Rev. 44, 767–784. http://dx.doi.org/10.1016/ j.rser.2015.01.029.
- Pérez-Díaz, J.I., Guisández, I., Chazarra, M., Helseth, A., 2020. Medium-term scheduling of a hydropower plant participating as a price-maker in the automatic frequency restoration reserve market. Electr. Power Syst. Res. 185, 106399. http://dx.doi.org/10.1016/j.epsr.2020.106399.
- Qadir, S.A., Al-Motairi, H., Tahir, F., Al-Fagih, L., 2021. Incentives and strategies for financing the renewable energy transition: A review. Energy Rep. 7, 3590–3606. http://dx.doi.org/10.1016/j.egyr.2021.06.041.

- Ramos, A., et al., 2022. openTEPES: open-source transmission and generation expansion planning. Institute for Research in Technology, Madrid. https://opentepes.readthedocs.io/en/latest/. (Accessed 31 Jan 2022)..
- REE Activities, 2020. REE. https://www.ree.es/es/actividades/operacion-delsistema-electrico/procedimientos-de-operacion (accessed Dec. 30, 2020).
- REE, 2009. Importancia del equipo generador hidroelectrico en la opercaion del sistema electrico. Instalaciones hidroelectricas estrategicas. [Online]. Available: https://www.chsegura.es/export/descargas/planificacionydma/ planificacion/docsdescarga/103_Aportaciones_al_EpTI_de_Red_Electrica_ Espanola_xREEx.pdf.
- Shaqsi, A.Z.A.L., Sopian, K., Al-Hinai, A., 2020. Review of energy storage services, applications, limitations, and benefits. Energy Rep. 6, 288–306. http://dx.doi. org/10.1016/j.egyr.2020.07.028.
- Tejada-Arango, D.A., Lumbreras, S., Sanchez-Martin, P., Ramos, A., 2020. Which unit-commitment formulation is best? A comparison framework. IEEE Trans. Power Syst. 35 (4), 2926–2936. http://dx.doi.org/10.1109/TPWRS.2019. 2962024.
- The European Commission, The European Commission, 2017. Establishing a guideline on electricity balancing. commission regulation (eu). p. 48, https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A32017R2195.
- Ventosa, M., Baillo, A., Ramos, A., Rivier, M., 2005. Electricity market modeling trends. Energy Policy 33 (7), 897–913. http://dx.doi.org/10.1016/j.enpol.2003. 10.013.
- Werlang, A., Cunha, G., Bastos, J., Serra, J., Barbosa, B., Barroso, L., 2021. Reliability metrics for generation planning and the role of regulation in the energy transition: Case studies of Brazil and Mexico. Energies 14 (21), 7428. http: //dx.doi.org/10.3390/en14217428.